\documentstyle[12pt]{article}

%\pdfcompresslevel=9
%\pdfoutput=1

\textwidth6.25in
\textheight8.5in
\oddsidemargin.25in
\topmargin0in

\pagestyle{empty}

\begin{document}

\begin{center}
     {\bf Complex Analysis,\ \ \ MA 648-1C \ \ \  }    Fall 2005
\end{center}

\noindent Class meets M,W,F 9:10-10:00am in Room EB 127
\vspace*{3mm}

\noindent{\bf Instructor:  }  Dr. Nikolai Chernov \\ {\bf Office:
} 492 A Campbell Hall, ph. 934-2154 or 934-8613 \\ {\bf Office
hours:}    Monday, Wednesday, Friday 9:15-10:00am, and by appointment\\
{\bf E-mail: } chernov@math.uab.edu\vspace*{3mm}

\noindent{\bf Text:}  J. Conway, {\em Functions of One Complex
Variable I}, 2nd Ed. \vspace*{3mm}\\
{\bf Syllabus:} Chapters 1--6. \vspace*{3mm}

\noindent{\bf Homework} problems will be assigned from the book on
a regular basis. The homework is due the next week after
assignment. Half credit is given for late homework. \vspace*{3mm}

\noindent{\bf Grading policy:}\vspace*{3mm}

\hspace*{14mm} Homework  \hspace*{\fill}      30 \%
\\ \hspace*{20mm} Midterm Test (tentatively: mid-October) \hspace*{\fill}  30  \% \\
\hspace*{20mm} Final Exam (Dec 14, 8:00-10:30)
\hspace*{\fill}  40 \%\vspace*{5mm}

\noindent This flyer and all homework assignments will be
available on the instructor's web page
$$
{\rm www.math.uab.edu/chernov/teach.html}
$$

\noindent All tests in this course are {\bf open-book} and
{\bf open-notes}.\\
\vspace*{3mm}

\end{document}
\end
